\chapter{Introduction}
\pagenumbering{arabic} \setcounter{page}{1}

A short paragraph introducing the topic the chapter examines.


\section{Background}

A number of pages about the background of the project.

\section{About this Thesis}
This is the report of \emph{\authorName}, submitted as part of the requirements for the degree of \degreeAward at the School of Computing and Digital Technology, Birmingham City University, Birmingham, UK..

A number of paragraphs detailing the main expectations of this body of work.


\section{Chapter List}
Provide a list of all the chapters within the report and a brief summary of the content. Ensure each summary avoids having a repetitive structure such as starting with ``This chapter deals with''.

\textbf{Chapter \ref{ch:usingLatex}} Using \LaTeX. This chapter
deals with how to use the \LaTeX \space system.

\textbf{Chapter \ref{ch:litReview}} Literature Review. This chapter
deals with $\ldots$.

\textbf{Chapter \ref{ch:Background}} Background Research. This chapter
deals with $\ldots$.

\textbf{Chapter \ref{ch:Design}} Design. This chapter
deals with $\ldots$.

\textbf{Chapter \ref{ch:Implementation}} Implementation. This chapter
deals with $\ldots$.

\textbf{Chapter \ref{ch:Evaluation}} Evaluation \& Testing. This chapter
deals with $\ldots$.

\textbf{Chapter \ref{ch:Conclusion}} Conclusion. The conclusions of the report are presented.


\section{Conclusion}
A short conclusion summarising the chapter.
